Author's response to reviews

Title: Postal survey methodology to assess patient satisfaction in a suburban emergency medical services system: an observational study.

Authors:

Aaron W Bernard (aaronwbernard@yahoo.com)
Christopher J Lindsell (christopher.lindsell@uc.edu)
Daniel A Handel (dan.handel@lycos.com)
Lindsey Collett (lcollett@zoomtown.com)
Paul Gallo (pgallo@readingohio.org)
Kevin D Kaiser (kkaiser@readingohio.org)
Donald Locasto (LOCASTD@UCMAIL.UC.EDU)

Version: 5 Date: 8 June 2007

Author's response to reviews:

Dear Editor,

Attached is the latest version of the manuscript with the requested formatting changes. The title was changed to sentence case, the authors qualifications were removed, underlining was removed, authors contributions now reflects the final accepted manuscript, and figures are uploaded separately.

Sincerely,
Aaron Bernard